Author’s response to reviews

Title: Cost-effectiveness of blended vs. face-to-face cognitive behavioural therapy for severe anxiety disorders: study protocol of a randomized controlled trial

Authors:
Geke Romijn (g.a.romijn@vu.nl)
Heleen Riper (h.riper@vu.nl)
Robin Kok (r.n.kok@vu.nl)
Tara Donker (t.donker@vu.nl)
Maartje Goorden (goorden@bm.eur.nl)
Leona Hakkaart-van Roijen (hakkaart@bm.eur.nl)
Lisa Kooistra (l.c.kooistra@vu.nl)
Anton van Balkom (t.vanbalkom@ggzingeest.nl)
Jeroen Koning (jeroenkoning@hotmail.com)

Version: 1 Date: 01 Dec 2015

Author’s response to reviews:

Dear reviewers,

In this revised submission all email addresses of all co-authors are included in the title page of the manuscript and the trial registration date had been adjusted. Further, the Authors Contributions' section had more detailed information about the individual contribution for each individual author listed on the manuscript.

Best wishes,

Geke Romijn